\begin{table}%t1
\caption{List of target {\emph{INTEGRAL}} sources observed with GMRT.}\label{t1}
%\centerline
{\begin{tabular}{lllllll}
\hline
\hline
Source           &Type        &Integral     &Variable             &\mbox{X-ray}     &\mbox{X-ray}    &GMRT         \\
                 &            &Pos. Unc     &100 s--1 ks             &Flux      &State    &detection    \\
                 &            &1.6$\sigma$  &IBIS/ISGRI           &(mCrab)   &during   &within \mbox{X-ray} \\
                 &            &             &                     &15--40~keV &GMRT obs.&3$\sigma$ error box\\
\hline
IGR~J06074$+$2205  &--            &2$'$           &Yes                 &15    &high soft&Yes\\
IGR~J15479$-$4529  &--            &2$'$           &Yes                 &2     &--        &Yes\\
IGR~J16316$-$4028  &--            &3$'$           &Yes                 &25    &--        &Yes\\
IGR~J16318$-$4848  &sgB(e)(HMXB) &4$''$        &Yes                 &50--100&high soft&No \\
IGR~J16320$-$4751  &HMXB(?)      &4$''$        &Yes                 &10--50 &--        &No \\
IGR~J16418$-$4532  &--            &2$'$           &--                   &2     &--        &No \\
IGR~J16479$-$4514  &             &3$'$           &Yes                 &12    &high soft&Yes\\
IGR~J17091$-$3624  &XB$^{1}$(BHC)&0.8$'$         &Yes                 &40    &low hard &Yes\\
                   &?            &               &                    &--     &--        &--  \\
IGR~J17391$-$3021  &Be/(NS)HMXB$^{2}$&1$''$    &Yes                 &150   &--        &No \\
                   &?            &Chandra        &                    &--     &         &   \\
IGR~J17544$-$2619  &HMXB(?)      &4$''$        &Yes                 &60    &high soft&No \\
IGR~J17597$-$2201  &Neutron Star, LMXB&2$'$      &Yes                 &5     &--        &No \\
IGR~J18027$-$1455  &Extragalactic$^{3}$&2$'$           &Yes                 &--     &high soft&Yes\\
IGR~J18325$-$0756  &--            &2--3$'$         &Yes                 &--     &-        &No \\
IGR~J18483$-$0311  &--            &2$'$           &Yes                 &10    &--        &No \\
IGR~J18539$+$0727  &XB$^{4}$     &3$'$           &Yes                 &20    &low hard &Yes\\
                   &(BHC)        &               &                    &--     &         &   \\
IGR~J19140$+$0951  &HMXB$^{5}$   &1.3$'$         &Yes                 &50$--$100&high soft&No\\
                   &(NS)        &               &                    &--     &         &   \\
IGR~J21247$+$5058  &Radio Galaxy &2$'$           &Yes                 &--     &low hard &Yes\\
\hline
\end{tabular}}
\par
\medskip
\par
{$^1$ Revnivtsev {et~al.} (\cite{Revnivtsev2003a}a), $^2$ Smith {et~al.} (\cite{Smith2003}), $^3$ Masetti {et~al.} (\cite{Masetti2004}),\\ $^4$
Revnivtsev {et~al.} (\cite{Revnivtsev2003b}b), $^5$ Rodriguez {et~al.} (\cite{Rodriguez2005}).}
\end{table}